Author's response to reviews

Title: Use of drug-eluting balloon coronary intervention prior to living donor kidney transplantation.

Authors:

Tobias Kammerer (tobias.kammerer@med.uni-muenchen.de)
Andres Beiras-Fernandez (andres.beiras@kgu.de)
Markus Rehm (markus.rehm@med.uni-muenchen.de)
Manfred Stangl (manfred.stangl@med.uni-muenchen.de)
Markus Guba (markus.guba@med.uni-muenchen.de)
Christian Kupatt (christian.kupatt@med.uni-muenchen.de)
Florian Weis (florian.weis@klinikum-ffb.de)

Version: 5
Date: 11 August 2014

Author's response to reviews:

The following items were modified:

1) Email addresses of all authors on the title page.
   All email addresses have been updated and inserted on the front page. Name of author Kupatt was changed to Kupatt-Jeremias. Hospital of author Florian Weis has changed to Clinic Fürstenfeldbruck. Email address of Andres Beiras-Fernandez has been updated.

2) The abstract in the manuscript was divided into sections “Background“, „Case presentation“ and „Conclusions“.

3) and 4) The sections „Patient’s consent“, „Competing interests“ and „Authors’ contributions“ were added to the manuscript.